



February 6, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 15674-review.doc).

Title: Metastatic pancreatic cancer: is there a light at the end of the tunnel?

Author: Vanja Vaccaro, Isabella Sperduti, Sabrina Vari, Emilio Bria, Davide Melisi, Carlo Garufi, Carmen Nuzzo, Aldo Scarpa, Giampaolo Tortora, Francesco Cognetti, Michele Reni, Michele Milella

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 15674

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

- (1) We edited row 6, page 6
- (2) We summarize the acronyms and abbreviations in a table
- (3) We split table 2 in two smaller table according to type of treatment
- (4) We shortened the text, eliminating repetitions and redundancies
- (5) A professional English language editor (native speaker of English) polished the manuscript

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

A handwritten signature in black ink that reads 'Vanja Vaccaro'.

Vanja Vaccaro, MD
Medical Oncology A
Regina Elena National Cancer Institute
Rome
00144 - Italy
Fax: +39-06-52665637
E-mail: vanja.vaccaro@hotmail.it